Author's response to reviews

Title: Involving deprived communities in improving the quality of primary care services: does participatory action research work?

Authors:

Peter G Cawston (p.cawston@nhs.net)
Stewart W Mercer (sm83z@clinmed.gla.ac.uk)
Rosaline S Barbour (r.barbour@dundee.ac.uk)

Version: 5 Date: 14 June 2007

Author's response to reviews:

Dear Editor
We have changed the competing interests statement as requested. We have removed box a1 and box 2 and simply inserted the contents (the questions asked) into the text of the methods and removed references to the boxes. We have added the tables to the end of the manuscript as requested. We have thus removed the additional files, except the two figures.